Dear Editors,



Please find enclosed the edited manuscript in Word format (file name: 3874-review.docx).

Title: Air embolism complicating gastrointestinal endoscopy: A systematic review

Author: Suman Donepudi, MD, Disaya Chavalitdhamrong, MD, Liping Pu, MS, and Peter V. Draganov, MD

Name of Journal: World Journal of Gastrointestinal Endoscopy

ESPS Manuscript NO: 3874

The manuscript has been improved according to the suggestions of reviewers:

- 1 Format has been updated.
- 2 Revision has been made according to the suggestions of the reviewer.
 - (1) We have corrected one sentence that has been typo.
- 3 References and typesetting were corrected.

Thank you again for publishing our manuscript in the World Journal of Gastrointestinal Endoscopy.

Sincerely yours,

Peter V. Draganov, MD University of Florida Room HD 602, PO Box 100214 Gainesville, FL 32610, USA

Fax: +1-352-392-3618

E-mail: dragapv@medicine.ufl.edu